Author's response to reviews

Title: Measuring the Barriers against Seeking Consultation for Urinary Incontinence among Middle Eastern Women

Authors:

Ahmed S. El-Azab (elazab@aun.edu.eg)
Omar M. Shaaban (omshaaban@yahoo.com)

Version: 2 Date: 23 June 2009

Author's response to reviews:

June 23, 2009

Editor in Chief, the BMC Women’s Health

Re: Measuring the Barriers against Seeking Consultation for Urinary Incontinence among Egyptian Women. 5101556642781896

Dear Editor – in - chief

Enclosed is our revised original manuscript entitled “Measuring the Barriers against Seeking Consultation for Urinary Incontinence among Egyptian Women” that we would like to submit for publication in the BMC Women’s Health. The authors have made all the corrections and comments made by the reviewers.

The comment made by Dr. Khunying Kobchitt Limpaphayom: “need to clarify the term UDI-6 , in the paper written , there is no place for clarification of this word. It did appear in the ref. # 2” …. We have added Urogenital Distress Inventory – short form (UDI-6) in page 4 to clarify the term

We are very grateful to the editorial board of BMC Women’s Health for revising our manuscript and considering it for publication in the BMC Women’s Health.

Cordially

Corresponding Author
Ahmed S. El-Azab, MD
Staff Urologist
Section of Female Urology and NeuroUrology
Urology Department
Assiut University Hospital
Assiut 71516
Egypt